%%    TEMPLATE for articles submitted to the full-tex econf proceedings
%%     
%%
%%     Please do not remove lines commented out with %+
%%           these are for the editors' use.
%%
%%     Questions?  Send email to :  mpeskin@slac.stanford.edu
%%   

\documentclass[12pt]{article}        
\usepackage{epsfig}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%  Do not change these:
\textwidth=6.0in  \textheight=8.25in

%%  Adjust these for your printer:
\leftmargin=-0.3in   \topmargin=-0.20in


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%  personal abbreviations and macros
%    the following package contains macros used in this document:
\input econfmacros.tex
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%  To include an item in the INDEX of the conference volume,
%           flag it with    \index{<item name>}
%  The use of this macro is illustrated in the text.
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\def\Title#1{\begin{center} {\Large {\bf #1} } \end{center}}

\begin{document}
\begin{thebibliography}{99}

%%
%%  bibliographic items can be constructed using the LaTeX format in SPIRES:
%%    see    http://www.slac.stanford.edu/spires/hep/latex.html
%%  SPIRES will also supply the CITATION line information; please include it.
%%

\bibitem{lhcb}
LHCb collaboration, A. A. Alves Jr. {\emph et al.}, JINST {\bf 3}, S08005 (2008).

\bibitem{buras}
A. J. Buras {\emph et al.}, JHEP 1010 (2010).

\bibitem{babu}
KS. Babu and C.F. Kolda, Phys. Rev. Lett. {\bf 84} 228 (2000).

\bibitem{cls}
T. Junk, Nucl. Instrum. Meth. {\bf 434} 435 (1999)\\
A. Read J. Phys. {\bf G28} 2693 (2002)

\bibitem{bsmm}
LHCb collaboration, Phys. Rev. Lett.{\bf 108} 231801 (2012).


\end{thebibliography}



\Title{\Bqmm\ searches at LHCb}

\bigskip\bigskip

%+\addtocontents{toc}{{\it D. Reggiano}}
%+\label{ReggianoStart}

\begin{raggedright}  

COSME ADROVER PACHECO\index{Adrover, C.} on behalf of the LHCb collaboration\\
{\it Centre de Physique des Particules de Marseille (CPPM)\\
Aix-Marseille Universit\'e, France}\\
\bigskip\bigskip
\end{raggedright}


 
\section{Introduction}
      
The rare decays \Bsmm\ and \Bdmm\ are benchmark channels to constrain models beyond the Standard Model (SM) with a larger Higgs sector. 
In the SM these processes are highly suppressed as they occur through loop processes. 
Together with an associated helicity suppression renders the branching fraction of these decays to be~\cite{buras}: $\mathcal{B}(B^{0}_{(s)} \to \mu^+ \mu^-) = 3.2 \pm 0.2 \times 10^{-9}$ and $\mathcal{B}(B^{0} \to \mu^+ \mu^-) = 0.10 \pm 0.01 \times 10^{-10}$. 

\vskip 2mm  
Any deviation from these well predicted branching fractions can lead to indications of physics beyond the SM. 
For instance, they can be significantly enhanced within Minimal Supersymmetric extensions of the SM (MSSM) due to contributions from
new processes or new heavy particles. In these scenarios, $\mathcal{B}$(\Bsmm) is proportional to $\tan^6 \beta$~\cite{babu}, where $\tan \beta$ is
the ratio of the vacuum expectation values of the two Higgs fields. 



\section{Branching fraction measurements of $B^{0}_{(s)} \to \mu^+ \mu^-$ at the LHCb}
The LHCb experiment~\cite{lhcb} has collected 1.0\,fb$^{-1}$ of data during the 2011 at a center-of-mass energy of $\sqrt{s}=7$\,TeV. 

The \Bqmm\ analysis starts with a loose and efficient selection of the \Bqmm\ candidates and control channels (\Bhh\ decays, where $h$ can be either a pion or a kaon, \BuJpsiK\ and \BsJpsiPhi\ decays). The main goal of this selection process is to reduce the background to manageable levels, while keeping the same efficiency in all aforementioned channels.

After the selection, \Bsmm\ candidates are classified according to their invariant mass and the output of a multivariate classifier (figure~\ref{fig:plot} on the left) combining
geometrical and kinematic information. This classification is performed in a binned 2D parameter space.
The signal expectation in each bin is calculated using data events such as \Bhh\ and \BuJpsiK\ decays.
The number of background is obtained after an interpolation from the invariant mass sidebands.

\vskip 2mm
The distribution of observed events (figure~\ref{fig:plot} on the right) is finally compared with the signal plus background and background only hypothesis using the CL$_S$ method~\cite{cls}.

The upper limits~\cite{bsmm} on the branching fractions are:
\begin{equation}
\mathcal{B}(B^{0}_{(s)} \to \mu^+ \mu^-) < 4.5 \times 10^{-9} ,
\end{equation}
\begin{equation}
\mathcal{B}(B^{0} \to \mu^+ \mu^-) < 1.0 \times 10^{-10}.
\end{equation}  

\begin{figure}[htb]
\begin{center}
\epsfig{file=bdt,height=2in}
\epsfig{file=events,height=2in}
\caption{On the left, the output of the multivariate classifier for signal \Bqmm\ events (black dots) and background (empty blue dots). On the right, distribution of the selected di-muon events in the \Bsmm\ mass window with BDT (classifier) requirement above 0.5, combinatorial background (light gray) and SM \Bsmm\ (gray).}
\label{fig:plot}
\end{center}
\end{figure}

The aforementioned limits are the most restrictive limits on these branching fractions obtained to date.






%\def\Discussion{
%\setlength{\parskip}{0.3cm}\setlength{\parindent}{0.0cm}
%     \bigskip\bigskip      {\Large {\bf Discussion}} \bigskip}
%\def\speaker#1{{\bf #1:}\ }
%\def\endDiscussion{}

%\Discussion

%\speaker{D. Giovanni (University of Seville)}  My analysis indicates that the
%recovery of the two gentlemen is due simply to their embrace of the masculine
%principle and has nothing to do with magnetism at all.  Could you comment on 
%this?

%\speaker{Reggiano} Professor Giovanni has discussed this hypothesis in several
%forums, but, I do not believe there is anything in print.  I understand that
%he is spending his time in other pursuits.

%\speaker{D. Anna (University of Seville)}  In fact, my colleague Giovanni 
%has expressed opposite opinions on this question at various times, depending
%on the audience.  All of these testosterone-based theories are, of course,
%nonsense.

%\endDiscussion
 %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%
%%   use this format to include an .eps figure into your paper
%%
%\begin{figure}[htb]
%\begin{center}
%\epsfig{file=magnet.eps,height=1.5in}
%\caption{Plan of the magnet used in the Mesmeric studies.}
%\label{fig:magnet}
%\end{center}
%\end{figure}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%





%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%
%%   use this format to include a LaTeX table  into your paper
%%
%\begin{table}[b]
%\begin{center}
%\begin{tabular}{l|ccc}  
%Patient &  Initial level($\mu$g/cc) &  w. Magnet &  
%w. Magnet and Sound \\ \hline
% Guglielmo B.  &   0.12     &     0.10      &     0.001  \\
% Ferrando di N. &  0.15     &     0.11      &  $< 0.0005$ \\ \hline
%\end{tabular}
%\caption{Blood cyanide levels for the two patients.}
%\label{tab:blood}
%\end{center}
%\end{table}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



%\bigskip
%I am grateful to Don Alfonso d'Alba for certain services essential to 
%this investigation.



\end{document}

